Author's response to reviews

Title: Hexane fraction of Ardisia crispa Thunb A.DC root inhibits inflammation-induced angiogenesis

Authors:

Dayang Erna Zulaikha Awang Hamsin (dayangerna@hotmail.com)
Roslida Abdul Hamid (roslida@medic.upm.edu.my)
Latifah Saiful Yazan (latifah@medic.upm.edu.my)
Che Norma Mat Taib (norma@medic.upm.edu.my)
Looi Ting Yeong (joyce_ylt86@hotmail.com)

Version: 6 Date: 21 December 2012

Author's response to reviews: see over
Dear Editor,

I herewith attached the copyedited manuscript upon the comment by the editor on 7th December 2012. The manuscript has been checked by Edanz Editing Service as further recommended. The comments and copyediting of the ms are attached in the tracked copy document. The revised manuscript is labelled as clearcopy document. The ms has undergone two rounds of copyediting by the respected editing service company. References are also reformatted according to the journal’s format.

Thanks

Regards

Roslida